Author’s response to reviews

Title: Applying the Analytic Hierarchy Process in healthcare research: A systematic literature review and evaluation of reporting

Authors:
Katharina Schmidt (ks@cherh.de)
Ines Aumann (ia@cherh.de)
Ines Hollander (ines.hollander@gmail.com)
Kathrin Damm (kd@cherh.de)
J.-Matthias von der Schulenburg (jms@cherh.de)

Version: 2 Date: 14 Dec 2015

Author’s response to reviews:

Dear Editor,

thank you very much for your comments. As there were no supplementary reviewer comments, it was easy for me to follow your technical remarks.

1. We include the PRISMA checklist and the Flow Chart in the supplementary files.

2. and 3. The annex and figures are removed from the manuscript and are uploaded in supplementary files. We are sorry that we did not noted that point in earlier versions.

4. All authors meet the ICMJE requirements as we state on page 16.